Author’s response to reviews

Title: Nurse managers' experience with ethical issues in Malaysia: A cross-sectional study

Authors:

Maizura binti Musa (maizuramusa@yahoo.com)
Md. Harun-Or-Rashid (mh_rashid67@yahoo.com)
Junichi Sakamoto (sakamjun@nagoya-u.ac.jp)

Version: 2 Date: 9 June 2011

Author’s response to reviews: see over
Dear editorial board of BMC Medical Ethics,

Please find enclosed the revised manuscript entitled “Nurse managers’ experience with ethical issues in Malaysia: A cross-sectional study”, by Maizura binti Musa et al., to be submitted as Original Research Article to BMC Medical Ethics for possible evaluation.

The reason for sending the revised manuscript is not due to the change of the contents, but because the initial manuscript contained red colored words in some pages, as below:

i. Page 9, paragraph 3, line 3;  
ii. Page 11, paragraph 2, line 4 and 5;  
iii. Page 13, paragraph 2, line 1;  
iv. Page 15, paragraph 1, line 2 and 3; and  
v. Page 16, paragraph 1, line 5

We have made the necessary correction and hoping that the editorial board and the reviewers will agree on the interest of this study.

Thank you.

Yours sincerely,

Maizura binti Musa,  
Corresponding author  
Email: maizuramusa@yahoo.com, phone/fax number +81-52-744-2444.